Author's response to reviews

Title: The effect of two different health messages on physical activity levels and health in sedentary, overweight middle aged women

Authors:

Sebely Pal (s.pal@curtin.edu.au)
Cheryl Cheng (c.cheng1@curtin.edu.au)
Suleen Ho (s.ho@curtin.edu.au)

Version: 4 Date: 10 March 2011

Author's response to reviews:

Formatted as requested